Author's response to reviews

Title: Increased Mortality among Survivors of Myocardial Infarction with Kidney Dysfunction: the Contribution of Gaps in the use of Guideline-Based Therapies

Authors:

Pamela N Peterson (pamela.peterson@uchsc.edu)
Amrut V Ambardekar (Amrut.Ambardekar@uchsc.edu)
Philip G Jones (pgjones@saint-lukes.org)
Harlan M Krumholz (Harlan.Krumholz@yale.edu)
Erik Schelbert (Schelberteb@nhlbi.nih.gov)
John A Spertus (spertusj@umkc.edu)
John S Rumsfeld (John.Rumsfeld@uchsc.edu)
Frederick A Masoudi (Fred.Masoudi@uchsc.edu)

Version: 5 Date: 6 July 2009

Author's response to reviews: see over
July 6, 2009

Editorial Board, BMC Cardiovascular Disorders

To Whom It May Concern:

Thank you for the opportunity to revise the manuscript 1381109276209660. We have made the requested changes to the Title Page, Abbreviations, and the figures. Attached, please find our revised manuscript, “Increased Mortality among Survivors of Myocardial Infarction with Renal Dysfunction: the Contribution of Gaps in the use of Guideline-Based Therapies.”

Please do not hesitate to contact me at (303) 436-5663 or Pamela.Peterson@ucdenver.edu with any questions.

Sincerely,

Pamela N. Peterson, MD MSPH
Assistant Professor of Medicine
Denver Health Medical Center
Mail Code 0969
777 Bannock St.
Denver, CO 80204
Phone: 303 436-5663
Fax: 303 436-7739
e-mail: Pamela.Peterson@ucdenver.edu